Author’s response to reviews

Title: Preparing for the unexpected: Special considerations and complications after sugammadex administration

Authors:

Hajime Iwasaki (iwasakih@asahikawa-med.ac.jp)
Johnathan Renew (Renew.Johnathan@mayo.edu)
Takayuki Kunisawa (taka.kunisawa@nifty.ne.jp)
Sorin Brull (SJBrull@me.com)

Version: 3 Date: 18 Sep 2017

Author’s response to reviews:

Dear Dr. Lingling Tian

We thank you for your comments. The original comments of the editor and our responses are as follows:

Editor Comments:

1 Figure 1

It is noted that figure 1 was modified from another source. Please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

It is also noticed the source cited here [4] is wrong. The author Iwasaki is not for this work.


We adapted the figure from Iwasaki et al. [7]. The figure is our original and therefore we think that we do not need permission for this figure. We have modified the cited number.
2 Figure 2

Please clarify whether figure 2 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

This figure is our original.

3 Clean copy

Please ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript (and please ignore any automated requests to show tracked changes). Should you wish to respond to these revision requests, please include the information in the designated input box only.

We will attach the clean copy of our manuscript.

Sincerely,

Hajime Iwasaki M.D., Ph.D.
Assistant professor
Anesthesiology and Critical Care Medicine
Asahikawa Medical University, Japan